JABES 26,1

76

Received 27 July 2018 Revised 29 October 2018 Accepted 28 November 2018

Asymmetric targeting of corporate cash holdings and financial constraints in Pakistani firms

Ghulam Ayehsa Siddiqua and Ajid ur Rehman
Faculty of Management Sciences, Riphah International University,
Islamabad, Pakistan, and
Shahzad Hussain

Faculty of Business and Technology, Foundation University, Islamabad, Pakistan

Abstract

Purpose – The purpose of this paper is to investigate the asymmetric adjustment of cash holdings in Pakistani firms for above and below target firms.

Design/methodology/approach – The study employs generalized method of moments (GMM) to investigate the adjustment of cash holdings.

Findings – The study found that the firms which hold cash above the optimal level of cash holdings have higher speed of adjustment than the firms which hold cash below the optimal level. Financially constrained (FC) firms also adjust their cash holdings faster than financially unconstrained (FUC) firms but high speed of downward adjustment does not remain persistent after financial constraints are controlled. Findings of this study reveal this asymmetric adjustment in above and below target firms and extend these results in FC and FUC Pakistani listed firms, respectively.

Research limitations/implications – The conclusion of this study has been derived under certain limitations. There is a vast space to extend this study in different dimensions. Firms operating in capital-intensive industries may provide different results for financial constraints because their policy designing would be quite different from other firms.

Originality/value — This study contributes to cash holdings research in Pakistan by exploring the adjustment behavior of cash holdings across Pakistani non-financial firms using econometric modeling. Downward adjustment rate is supposed to be higher than upward adjustment rate and this rate is tested using dynamic panel data model. Similarly, it is inferred that this relationship holds for above target firms even after including the financial constraints in the presented model.

Keywords Cash holdings, Adjustment rate, Financial constraints, Pakistani firms, Upward and downward adjustment

Paper type Research paper

1. Introduction

The cash holding behavior of firms has obtained a great deal of consideration in finance literature after the contribution of Miller and Orr (1966) and the initial work of Modigliani and Miller (1958). However, the latter suggested that firms can easily secure funds in frictionless markets and that there is no need to hoard cash for future liquidity matters. Practically, there is no existence of frictionless capital markets and firms cannot always collect as many funds

© Ghulam Ayehsa Siddiqua, Ajid ur Rehman and Shahzad Hussain. Published in *Journal of Asian Business and Economic Studies*. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/licences/by/4.0/legalcode

The authors are extremely grateful to the valuable comments of the anonymous reviewers. Their suggestions have added great value to the manuscript.



Journal of Asian Business and Economic Studies Vol. 26 No. 1, 2019 pp. 76-97 Emerald Publishing Limited 2515-964X DOI 10.1108/JABES-07-2018-0056 as they need so they have to move toward external sources of raising funds. Why are the firms Corporate cash always in need of holding cash? Does the optimal level of cash holdings exist? Do firms with different organizational hierarchies hold a different amount of cash? To answer these fundamental questions, a number of researchers have strived to draw a clear picture of the cash holding decisions made by the firms, Keynes (1936) emphasized that cash acts as a safety measure against unpredicted contingencies. After three decades, a tradeoff model for determining a firm's optimal cash level was presented by Miller and Orr (1966) and this model discusses the idea of making a tradeoff between costs and benefits of cash holdings. Contrarily, Myers (1984) suggested pecking order theory and argued that an optimal or target level of cash does not exist and a firm only tries to minimize information asymmetry while accessing the costs of external financing. Under this argument, firm first use their retained earnings to finance their investment projects, then obtain debt and at last, they use their equity in their investments.

It is clear that firms do not have any optimal level of cash rather cash is simply used as a buffer between investment needs and retained earnings. Alternatively, Jensen (1986) presented the theory of free cash flow postulating when managers act for their own self-interest instead of striving for the value maximization of their firm, they may commit a breach of their fiduciary obligations toward shareholders. To understand the relationship between managers and shareholders, such kind of agency problems must be taken into account. Free cash flow theory holds that managers hold cash to exacerbate their arbitrary power over the investments decisions made by the firm.

However, holding cash has its benefits and costs. The basic purpose of hoarding cash includes a reduction in the chances of financial shocks (John, 1993), minimizing transaction costs (Keynes, 1936), circumventing external sources of financing and allowing the investment projects to perform efficiently in the presence of financial constraints (Denis and Sibilkov, 2010). Holding cash in a firm's reserves acts as a buffer against future financial shocks and firms tend to accumulate cash to cope with the financial crisis likely to occur in coming years. Holding cash also minimizes transaction cost of liquidating assets or costs associated with raising external finance (Mulligan, 1997).

However, accumulating huge volume of cash leads to double taxation especially for multinational firms that pay taxes in host country and are also subject to tax payments when repatriating foreign income to their home country (Foley et al., 2007), agency costs incurred due to conflicts between managers (agents of shareholders) and shareholders (Jensen, 1986; Harford, Li and Zhao, 2008) and opportunity cost (Uyar and Kuzey, 2014). Holding a large amount of cash may lead toward inefficiency. That is, the firm may lose certain valuable investment prospects. Firms hold cash for transaction motive, precautionary motive, agency motive and for tax motive as well. Pecking order theory suggests that firms tend to rely on internal financing more than external financing while making their investment decisions (Myers, 1984) and on the other side, agency theory (Jensen, 1986) points out a flaw, that is, when managers have excess cash, then they do not go for external sources of finance, they carry out such investment projects that may have even a negative net present value and at last, shareholders are adversely affected.

Broadly speaking, pecking order theory and agency theory do not sufficiently address the adjustment of cash holdings. So, a better explanation can be given by tradeoff theory which provides a balance between the benefits and costs which are associated with any given level of cash. An optimal or target level of cash is well determined by tradeoff theory and firms try to adjust cash to the optimal level in case of any cash deviations. This argument is relevant to hold that firms are active in rebalancing their cash holdings to the optimal level.

Numerous prior studies support the notion that an optimal or target level of cash holding exists for the firm (Opler et al., 1999; Ozkan and Ozkan, 2004; Bates et al., 2009; Rehman and



Wang, 2015). Although there is ample research material on adjustment of optimal or target level of cash holdings and adjustment rate, a very little work has been done on the asymmetric adjustment (high and low cash regimes from the optimal or target level) of corporate cash holdings in the particular context of Pakistan as most of the studies emphasized on cash holdings and adjustment rate of the firms operating in developed countries. No empirical evidence exists so far in the particular context of Pakistan which addresses the optimal level of cash holdings and adjustment rate of corporate cash holdings in Pakistani firms. Furthermore, the research is lacking in the strand of above and below target firms – how firms adjust their cash policy when the cash holdings are above or below the optimal level. Azam and Shah (2011) found that there are more financial constraints faced by Pakistani firms than the firms operating in the developed world. These constraints include high dividend payout ratio which restricts the firms to invest in future projects; firm's age which explains that older firms tend to spend less on investment as compared to the younger firms; and uncertainty which hinders fixed investment. Firm size. earnings and energy crisis are some other important constraints which need the attention of researchers. They further investigated the underlying relationship between a firm's level of investment and the firm size, age of the firm and its dividend payout ratio. Their findings revealed a positive linkage between investment and firm size and a negative association between investment, firm's age and dividend payout ratio. Consequently, firm's age and dividend payout ratio have been attributed to financial constraints.

This research is significant in certain strands. This study intends to contribute to existing literature by exploring cash management and adjustment of cash holdings in publicly listed non-financial firms. Furthermore, this research makes a contribution to literature because it is exploring the determinants of corporate cash holdings in Pakistan where the financial structure of firms is quite different from the firms operating in developed countries. The study intends to provide practicable insights and facts that may help to determine the asymmetric adjustment of cash holdings to help non-financial companies of Pakistan in their future investment and growth decisions and to understand the dynamics of optimal cash policy.

The rest of the study is structured as follows: Section 2 gives a brief review of the literature. Section 3 presents the data, methodology and empirical models. Section 4 deals with empirical results and Section 5 concludes the paper.

2. Literature review

There are a large number of prior studies about corporation's cash management policies and these studies suggest that firms normally accumulate large amounts of cash for precautionary motives (Opler *et al.*, 1999; Mikkelson and Partch, 2003), for efficient management of transactions (Mulligan, 1997), for payment of double taxes, i.e. multinational firms which are subject to tax payments both in host country and in home country as well (Foley *et al.*, 2007) and to reduce agency problems (Jensen, 1986; Harford, Mansi and Maxwell, 2008; Nikolov and Whited, 2014). Dittmar *et al.* (2003) identified two types of costs associated with holding cash. First, cost-of-carry and agency cost. They further documented two motives which stemmed the benefits and advantages of holding cash. In the first place, the transaction cost motive of holding cash states that firms hold more cash during the periods when opportunity costs and the costs associated with raising cash are relatively higher. Second, the precautionary or preventive motive of holding cash stems from an examination of the effect of asymmetric information on fund-raising ability of a firm.

According to the financing hierarchy (Myers, 1984), there is not any target level of cash and likewise, there is no optimal level of debt. But Martínez-Sola *et al.* (2013) and Jarrow *et al.* (2018) reported that there exists an optimal level of cash which maximizes the value of a firm and any divergence from the optimal level decreases firm value. The tradeoff theory

maintains a positive association between cash level and investment made for capital Corporate cash expenditure while financing hierarchy holds the opposite relationship between the two (Dittmar et al., 2003). Similarly, there exists an optimal level for debt or leverage which the firms obtain after making a tradeoff between benefits and costs of obtaining debt and any deviation from that optimal level may lead firms to move toward a new leverage target (Denis and McKeon, 2012). Denis (2011) held that as leverage ratios may substantially deviate from their target level, the managers do not set leverage levels as their first-order concern for capital structure decisions. Surplus leverage due to increase in initial leverage level builds cash reserves for firms. Furthermore, they suggested that during the time of shortage of cash and liquidity crisis, firms are active in taking more and more debt even when they are above their target or optimal debt level and likewise, during the time of surplus in financial resources they payout debt to reduce their leverage level even when they are already below their target level of leverage.

In capital markets where there is an ease of access to the fund providers and funds can immediately be raised, firms tend to keep less liquid assets in their reserves. In countries, where there is the least protection of investor's rights, companies hold twice much cash as companies in countries where investor rights are well protected. In this situation, investors cannot forbid managers to hold excessive cash. Financial instruments in a firm's portfolio also lessen cash hoarding because these instruments can easily be used for raising capital and for hedging as well. Furthermore, large amounts of cash are mostly held by the companies that are exposed to greater investment horizons and they hoard cash to avoid opportunity cost and a shortage of cash in case of optimal investment opportunity arousal. Precautionary motive of holding cash suggests that a firm's risk of refinancing also affects its level of cash holdings because firms hold huge amounts of cash to avoid refinancing risk and to save more cash resulting from free cash flows available to the finance providers (Harford et al., 2014; Xie et al., 2017).

In case of adjustment speed of cash holdings, different researchers hold different opinions. Chang et al. (2017) argued that firms have different adjustment costs so they follow different paths to reach their optimal level of cash. Furthermore, they hold that there is always an optimal level of cash and when the cash level deviates from the upper or lower cash regime then systematic adjustment of cash occurs. In this way, the benefits of cash level adjustment become higher than the costs. Jiang and Lie (2016) also examined the speed of adjustment of corporate cash holdings and they maintained that firms having higher levels of cash reserves have a higher speed of adjustment than the firms facing cash deficiency.

While addressing a firm's asymmetric adjustment, a firm can make loan payments and dividend payments when its level of cash holdings is above target or optimal level and by making such payments, it can bring its level of cash holdings down to the target or optimal level (Venkiteshwaran, 2011). This argument can be made by intuition and clue. Contrary to this argument, a firm cuts its investment, raises funds from external sources and slashes its payouts when its cash level is below the optimal level (Venkiteshwaran, 2011; García-Teruel and Martínez-Solano, 2008). Rehman et al. (2016) found a higher speed of downward adjustment of cash holdings than upward adjustment and this tendency is due to the reason that there are more alternatives available to the firms to bring their level of cash holdings down to the optimal or target level and lower costs associated with downward adjustment of cash holdings. So it can be suggested that it is far more convenient to bring the firm's cash holdings down to the optimal level when the level of cash holdings is above the optimal or target level than to bring the level of cash holdings up when it is below the optimal or target level during the time of uncertainty and crisis. Above arguments provide a base for the development of following hypothesis:

H1. Downward adjustment rate of corporate cash holdings toward an optimal level is higher than the upward adjustment rate.



JABES 26.1

80

2.1 Financial constraints and corporate cash holdings

Financial constraints have a different approach to explain a firm's cash holding tendency. Firms with a higher return on assets and firms which are paying the dividend can easily raise external finance and they hold less cash in their reserves (Chen *et al.*, 2017). Financially constrained (FC) firms are those which are not paying dividends and financially unconstrained (FUC) firms pay dividends (Chen *et al.*, 2017; Lozano and Durán, 2017) and FC firms hoard cash to deal with volatility in cash flows while FUC firms are not affected by this kind of volatility (Rehman *et al.*, 2016).

Almeida *et al.* (2004) argued that FC firms must adopt a different approach to cash saving and the approach should be a systematic propensity toward cash hoarding. FC firms are not required to adopt this approach. They further proposed that cash flows sensitivity toward cash holdings has a positive sign in case of FC firms and it is insignificant and negative for FUC firms. These findings support the opinion that FC firms have higher levels of cash than FUC firms. FC firms have higher levels of cash holdings as a result of higher investment yields and higher value of an investment (Denis and Sibilkov, 2009). During the times of cash crunches and less liquidity, firms normally cut their investment in research and development and technology (Campello *et al.*, 2010), and firms also tend to reduce their cash savings and dividends during such crisis. Assets liquidation can easily be made by FC firms at the time of liquidity crisis and the shortage of cash. Further linking this up to financial flexibility, marginal costs of excess cash and dynamics of capital structure, the ability to raise debt has a low transaction cost, meaning raising debt today to fund investment and subsequently seeking to pay off debt today, so that firm can raise more debt today or in future if needed (DeAngelo *et al.*, 2011). Based upon the above discussion, it can be hypothesized that:

H2. FUC firms have higher adjustment rate of cash holdings than FC firms.

Financially flexible firms tend to access low-cost external finance to timely respond sudden cash flow volatility and an unexpected increase in growth opportunities for value maximization (Denis, 2011); however, a firm's financial policy does not solely depend upon financial flexibility (Graham and Harvey, 2001). Financial constraints may restrict firms to avoid certain profitable projects so FC firms devise their cash policies with financial flexibility in order to cope with scarcity of financial resources during the periods of uncertainty and high cost of external finance and high uncertainty in growth opportunities lead firms to stockpile cash through low-equity payouts (Denis, 2011). For firms to be financially flexible, their unused debt capacity should be an important source of their capital structure. Gamba and Triantis (2008) reported that in case of higher debt costs, firms have a tendency to hoard more cash. They further argue that in time of low profitability, firms tend to reduce their debt burden, to avoid the triggering of any financial distress thus compelling firms to reduce their payout for debt issuance of higher costs.

Firms working with high market imperfections and with higher investment needs tend to keep large cash reserves in order to cope with liquidity crunches because market frictions restrict their investment ability (Almeida *et al.*, 2004). Furthermore, they argued that FUC firms are less prone to volatility in cash flows than FC firms. Rehman *et al.* (2016) incorporated financial constraints like Altman's Z score (based upon leverage, liquidity and profitability), SA1 and SA2 index (based upon size and age of the firm) in their research model and found that FUC firms have a higher speed of adjustment than FC firms. They also provided an argument that higher downward speed of adjustment toward the optimal or target level of cash is persistent even after the financial constraints are controlled. These arguments provide a base for the development of following hypothesis:

H3. Higher downward adjustment rate of corporate cash holdings persists even after financial constraints are controlled.



2.2 Determinants of corporate cash holdings

Opler et al. (1999) suggested several determinants of cash holdings and this study follows those determinants to be substantially incorporated in underlying regression models. The section below gives a brief summary of the relationship between cash holdings and the proposed determinants of cash holdings. We include capital expenditure, leverage, firm size, growth opportunities, net working capital and operating cash flows as the control variables.

3. Methodology

3.1 Data and source

We have used a sample set of 200 non-financial firms listed on Pakistan Stock Exchange over a ten-year period (2006–2016). Data are collected from www.psx.com.pk, www. businessrecorder.com, www.investing.com, annual reports of firms and BVD OSIRIS. Firms are assigned numbers ranging from 1 to 200 and then data have been split into two subcategories, i.e., firms which hold cash above the optimal or target level and firms which hold cash below the optimal or target level. The subsamples of data into above target firms and below target firms are based upon a technique borrowed from prior studies of capital structure (Hovakimian *et al.*, 2001; Drobetz and Wanzenried, 2006). First, cash holdings are estimated using pooled OLS estimation technique. Fitted values are estimated and then subtracted from the actual cash values. If the difference is positive it accounts for above target firms and negative values account for below target level firms. Data have been winsorized at the 5 percent level for limiting the extreme values and to reduce the effect of spurious outliers (Table I).

S. No.	Name of industry	No. of firms
1	Automobile manufacturer	6
2	Automobile add-ons	6
3	Electric products	2
4	Cement manufacturer	15
5	Chemical producers	15
6	Construction materials	4
7	Engineering	9
8	Fertilizer	4
9	Food and personal care	10
10	Glass and ceramics	6
11	Leather and tanneries	3
12	Miscellaneous	3
13	Oil and gas	9
14	Paper and board	7
15	Pharmaceuticals	5
16	Power generation	9
17	Refinery	2
18	Sugar	17
19	Synthetic and rayon	2
20	Technology	5
21	Textile	51
22	Tobacco	2
23	Transport	4
24	Woolen	4
Total	***************************************	200

Table I. Distribution of firms across industries



3.2 Variables description

3.2.1 Dependent variable. The dependent variable in this study is cash holdings. Cash holdings are the assets that a firm holds in the form of ready cash. The value of cash and cash equivalents has been taken from the annual reports of the firm and the online resources mentioned above. Cash is measured by dividing the cash holdings of a firm by its total assets.

3.2.2 Independent variables. Independent variables in this study are capital expenditure, leverage, growth opportunities, firm size, cash flow, net working capital and financial constraints. Capital expenditure is the amount of money spent on the acquisition of fixed assets. It is measured by dividing the value of fixed assets by total assets. Leverage refers to the investment of borrowed money. It is measured by dividing a firm's total debt by its total assets. Growth opportunities are the prospects for the firms to invest in projects which yield profits. These are measured by taking the ratio of the market value of equity and the book value of equity or it means to divide market value of equity with book value of equity. Firm size is the optimal size of a firm in a given industry at a given time which leads to low per unit cost of production. It is measured by taking the natural logarithm of a firm's total assets or its sales. Cash flow is the total amount of money coming in and going out of a business and it particularly affects liquidity. Cash flow is measured by dividing the net operating cash flows of a firm with total assets. Net working capital is the sum of all the liquid assets of a firm. It is measured by subtracting accounts payable from the sum of accounts receivables and inventories and dividing the resulting by total assets for scaling purpose.

3.3 Research model

First, we have developed a static model following Opler et al. (1999) and the model is as follows:

$$CASH_{it}^* = \beta_0 + \beta_1 NWC_{it} + \beta_2 SIZE_{it} + \beta_3 MTB_{it} + \beta_4 LEV_{it} + \beta_5 CAPEX_{it} + \beta_6 OCF_{it} + e_{it}.$$
 (1)

In Equation (1), $CASH_{it}^*$ refers to cash and cash equivalents held by firm i at time t. The star denotes that it is the optimal value or equilibrium represented by the fitted line of this equation. β_0 is the intercept. NWC_{it} represents net working capital employed by firm i at time t, measured by taking the difference of current assets and current liabilities. $SIZE_{it}$ is actually the firm's size which is measured by taking the natural log of total assets held by a firm. BMR_{it} depicts firm's book-to-market ratio used to measure the growth opportunities of firm i at time t. LEV_{it} stands for leverage of firm i at time t, measured by dividing total liabilities with total assets. $CAPEX_{it}$ is the ratio of firm's total capital expenditure to firm's total assets. OCF_{it} is the net operating cash flows of firm i at time t. e_{it} is the random error term.

The adjustment of cash holdings of a firm to a target or optimal level is not immediate and it has its associated costs as this adjustment takes place through a partial adjustment process. So the relationship given below holds the current cash holdings and cash holdings at t-1:

$$CASH_{it} - CASH_{it-1} = \gamma \left(CASH_{it}^* - CASH_{it-1} \right) + \delta_{it}. \tag{2}$$

$$CASH_{it} = \beta_0 \gamma + (1 - \gamma)CASH_{t-1} + \gamma \beta NWC_{it} + \gamma \beta_2 SIZE_{it} + \gamma \beta_3 BMR_{it}$$
$$+ \gamma \beta_4 LEV_{it} + \gamma \beta_5 CAPEX_{it} + \gamma \beta_6 OCF_{it} + \eta_i + \lambda_t + v_{it}.$$
(3)

In Equation (2), $CASH_{it}^*$ represents the cash level of firm i at time t and $CASH_{it-1}$ is the firm i's cash level at time t-1. $CASH_{it}^*$ denotes the target or optimal level of cash holdings of firm i at time t. γ denotes the coefficient of adjustment and its values range between 0 and 1. If $\gamma = 0$, it means that a firm will remain in its current cash position and if $\gamma = 1$, the firms

will tend to achieve an optimal or target level of cash holdings. By putting the value of Corporate cash $CASH_{it}^*$ from Equation (1) into Equation (2), we get the following equation.

In Equation (3), η_i corresponds to firm-specific effects and λ_t represents time-specific effects. By simplification of Equation (3), we have obtained the following equation:

$$CASH_{it} = \beta_0 \gamma + \rho CASH_{t-1} + \delta_1 NWC_{it} + \delta_2 SIZE_{it} + \delta_3 MTB_{it} + \delta_4 LEV_{it} + \delta_5 CAPEX_{it}$$

$$+\delta_6 OCF_{it} + \delta_7 CP_{it} + \delta_8 PP_{it} + \delta_9 LIQ_{it} + \delta_{10} TANG_{it} + \delta_{11} CV_{it} + \eta_i + \lambda_t v_{it}. \tag{4}$$

In Equation (4), $\alpha = \gamma \beta_0$; $\rho = (1-\gamma)$; $\delta_k = \gamma \beta_k$; and $\lambda_t v_{it} = \gamma e_{it}$. The use of OLS to estimate Equation (4) will lead to inconsistency because there is a problem of endogeneity between cash holdings and firm's adjustment toward the optimal level of cash. Hence, two-step generalized method of moments (GMM) estimator will be used to resolve the issue of endogeneity and to estimate Equation (4). The reason for selecting two-step GMM is that it is more efficient than one-step GMM.

This study has estimated the equation through GMM (Arellano and Bond, 1991). One of the reasons to estimate our equation through GMM is addressing the issue of endogeneity. In post-estimation test, we estimated the Sargan test value and Abond test for the presence of second-order autocorrelation. The p-values for both Sargan and Abond tests are used for the validity of these tests. Then we divided the firms into above and below target firms by estimating the equation through pooled OLS and subtracting the fitted values from actual values of cash holdings as done in various capital structure studies and more recently by Rehman et al. (2016). Furthermore, we included the financial constraints in our model and re-estimated the equation for above and below target firms to control for the financial constraints.

3.4 Measurement of financial constraints

Financial constraints are measured by using two methods.

3.4.1 Altman's Z score. First, Altman's Z score model is used in the study to identify financially flexible firms. This model was proposed by Bancel and Mittoo (2011). It captures some unique variables and is based upon liquidity ratios, profitability ratios and leverage ratios (i.e. debt to equity ratio):

$$Z = 1.2X_1 + 1.4X_2 + 3.3X_3 + 0.6X_4 + 0.999X_5$$

where X_1 is the cash ratio minus trade payables ratio. It is used to measure liquidity of firm. X_2 is the retained earnings divided by total assets; retained earnings are profits kept for reinvestment in business. X_3 is the earnings before interest and taxes divided by total assets. X_4 is the book value of equity divided by book value of total liabilities. X_5 is the sales divided by total assets.

The result or score is divided into three quartiles where the highest quartile represents those firms which are FUC and lowest quartile corresponds to firms which are FC.

3.4.2 SA index. SA index was proposed by Hadlock and Pierce (2010). It describes that firm's external factors are important to measure its financial constraints. SA index comprises size and age of the firm. Less constrained firms have high SA score and inverse will be the case for FC firms. In SA index, firm size is measured by taking the natural logarithm of its total assets or sales. Age of firm is calculated from the time of its listing:

$$SA1 = -0.737 Assets + 0.043 Assets^2 - 0.040 Firm's age.$$

SA2 = -0.737 Sales +0.043 Sales $^2 -0.040$ Firm's age.

After the calculation of SA1 and SA2, results are divided into three quartiles where quartile 3 corresponds to FC firms and quartile 1 represents the firms which are FUC.

3.5 Distribution of firms into above and below target level

Data have been split into two subsamples: firms which hold cash above the optimal level and firms which hold cash below the optimal level. This idea of categorizing firms into above and below target firms is adopted from Rehman *et al.* (2016), Hovakimian *et al.* (2001) and Drobetz and Wanzenried (2006). Firms estimate a target or optimal level of cash holdings after making a tradeoff between the costs and benefits associated with holding more cash. First, the model is estimated by simple OLS regression which gives results comprising fitted values of regression. The resulting values of regressed fitted line represent the optimal level of cash. The values of fitted line are subtracted from the actual value of dependent variable (cash) and if the result is a positive number, it means that actual value is higher than the estimated value and the firm is above the optimal or target level of corporate cash holdings. Inversely, if the answer is a negative value, it means the firm is below the optimal or target level of cash holdings.

4. Discussion of results

4.1 Descriptive statistics

Table II comprises descriptive statistics for overall firms, and a representation of their number of observations, mean and standard deviation. The mean value (average value) for cash is 0.07 with the standard deviation of 0.14. The average value for firm size is 4.64 with the standard deviation of 0.8. For leverage, the average value is 1.16 and its standard deviation is 0.85. Operating cash flow has a mean value of 0.06 and standard deviation of 0.26. Mean value for growth is 2.10 with a standard deviation of 3.45. Average value for net working capital is -0.03 with a standard deviation of 1.77. The mean value for capital expenditure is 0.55 with a standard deviation of 0.43. Altman's Z score's mean value is 1.39 with a standard deviation of 1.5.

Table III corresponds to descriptive statistics for firms above the target level of corporate cash holdings and firms below the target level of corporate cash holdings. For determination of optimal or target level of cash, fitted value of OLS regression has been subtracted from actual cash values. The resulting values are both positive and negative where positive values correspond to those firms which have cash holdings above the optimal or target level and negative values represent the firms which have cash holdings below the optimal or target level of cash. In Table III, mean value of cash for above target firms is much higher than below target firms. Mean value of operating cash flow is also higher for above target

Variable	Obs	Mean	SD
CASH	1,924	0.07	0.14
SIZE	1,924	4.64	0.8
LEV	1,924	1.16	0.85
OCF	1,924	0.06	0.26
GROW	1,923	2.10	3.45
NWC	1,924	-0.03	1.77
CAPEX	1,924	0.55	0.43
ZSCORE	1,921	1.39	1.5

Table II.Descriptive statistics for overall firms

Notes: Obs, observations; *CASH*, cash (dependent variable); *SIZE*, firm size; *LEV*, leverage; *OCF*, operating cash flow; *GROW*, growth opportunities; *NWC*, net working capital; *CAPEX*, capital expenditure; *ZSCORE*, Altman's *Z* score



		Above target		Below target			
Variable	Obs	Mean	SD	Obs	Mean	SD	
CASH	507	0.22	0.21	1,417	0.01	0.02	
SIZE	507	4.78	0.9	1,417	4.59	0.75	
LEV	507	1.1	1.06	1,417	1.18	0.76	
OCF	507	0.1	0.18	1,417	0.05	0.28	
GROW	506	3.40	4.1	1,417	2.51	3.2	
NWC	507	-0.07	3.4	1,417	-0.02	0.39	
CAPEX	507	0.52	0.73	1,417	0.57	0.23	
ZSCORE	506	2.07	1.52	1,415	1.14	1.42	

Corporate cash holdings

85

Notes: Obs, observations; CASH, cash (dependent variable); SIZE, firm size; LEV, leverage; OCF, operating cash flow; GROW, growth opportunities; NWC, net working capital; CAPEX, capital expenditure; ZSCORE, Altman's Z score

Table III. Descriptive statistics for above and below target firms

firms than for below target firms suggesting that above target firms tend to keep more cash to cope with liquidity crunches and financial distress. For leverage, mean value for both above and below target firms is not significantly different and it is slightly higher for below target firms which suggests that below target firms keep large amount of debts to deal with liquidity shortage. For growth opportunities, mean value is higher for above target firms than below target firms which means that firms try to hold more cash to finance higher growth opportunities. Mean value of net working capital is negative for both above and below target firms which indicates a large incurrence of current liabilities and a decrease in current assets. Capital expenditure is higher for below target firms than above target firms indicating that there are lower amounts of free cash flows to equity holders in below target firms and higher amounts of free cash flows are available to finance providers in above target firms.

4.2 Correlation matrix

Table IV represents correlation between all the variables of study. The last column corresponds to the variance inflation factor (VIF). To prove for the absence of multicollinearity, there should be no correlation between independent variable and the values of VIF must be less than 5. All the values of correlation matrix are within acceptable limits which correspond to the notion that there is no severe issue of correlation among independent variables. Furthermore, all values of VIF are also within the acceptable range (below 5). These two instances confirm the absence of multicollinearity between independent variables of the study.

4.3 Adjustment speed of overall firms

Arellano and Bond dynamic panel data model (GMM) is used to estimate Equation (4). Table V corresponds to the results of panel data regression for overall firms and the results are derived from applying GMM technique. In Table V, the value of coefficient is positive and statistically significant for lagged cash variable CASH (L1) where the value of coefficient is 0.583 and value of *t*-test is 22.67. It indicates that Pakistani firms follow the optimal or target level of cash holdings according to the tradeoff theory to keep a balance between costs and benefits of financing with debt and equity. The adjustment speed is calculated by subtracting the value coefficient of lagged cash variable from one.

The adjustment speed of overall firms is 0.417 (1-0.583) which is the indication of robustness of the results because the value of adjustment parameter ranges between 0 and 1. As the coefficient for the lagged value of cash is positive as well as statistically significant, it indicates that there is a partial adjustment policy followed by Pakistani firms toward the



CASH SIZE LEV OCF**GROW** NWC **CAPEX** VIF CASH1 SIZE 0.05 1 1.28 LEV-0.241.18 -0.131 OCF-0.091 1.08 0.10 0.01 GROW0.14 -0.130.14 1.06 0.11 1 NWC-0.090.07 0.05 -0.060.01 1 1.03 **CAPEX** -0.19-0.040.38 -0.01-0.020.006 1.02

86

Notes: *CASH* is the dependent variable cash which is measured through dividing the cash holdings of a firm by total assets. *SIZE* is the independent variable firm size which is the optimal size of a firm in a given industry and it is measured by taking natural logarithm of a firm's total assets. *LEV* is the independent variable leverage which refers to the investment of borrowed money and is measurement made by dividing total debt of firm by its total assets. *OCF* is the independent variable operating cash flow which is total amount of cash coming in and going out in a business and it is measured by dividing net operating cash flows of a firm with total assets. *GROW* is the independent variable growth which are prospects for a firm to invest in profitable projects. Their measurement is made by taking the ratio of market value of equity and book value of equity or in simple words, to divide market value of equity with book value of equity. *NWC* is net working capital that is the sum of all liquid assets and it is measured by subtracting current liabilities from current assets and then dividing the resulting figure with total assets. *CAPEX* is the independent variable capital expenditure which refers to the amount of money spent to acquire fixed assets and it is measured by dividing the amount of fixed assets by total assets. *VIF* is variance inflation factor

Table IV. Correlation matrix

Variables	Coef.	SE	t-test	<i>p</i> > <i>t</i>
CASH				
CASH (L1)	0.583	0.025	22.67	0.000
SIZE	0.006	0.015	0.44	0.66
LEV	0.01	0.003	3.04	0.002
OCF	0.141	0.021	6.68	0.001
GROW	0.001	0.005	1.41	0.16
NWC	-0.001	0.028	-0.24	0.812
CAPEX	0.019	0.014	1.41	0.159
_CONS	-0.04	0.077	-0.64	0.523
Adj. rate	0.417			
No. of groups	200			
No. of instruments	52			
Sargan test	0.329			
Abond test	0.385			

Table V.Dynamic panel data regression results for overall firms (GMM)

Notes: GMM is Arellano and Bond estimation. *t*-test values are given in the table. *CASH* is measured by dividing the cash holdings of a firm by total assets. CASH (L1) is the lagged cash variable. *SIZE* is measured by taking natural logarithm of a firm's total assets. *LEV* is measured by dividing total debt of firm by its total assets. *OCF* is measured by dividing net operating cash flows of a firm with total assets. *GROW* is the independent variable growth opportunities and measured by taking the ratio of market value of equity and book value of equity. *NWC* is measured by subtracting current liabilities from current assets and then dividing the resulting figure with total assets. *CAPEX* is measured by dividing the amount of fixed assets by total assets

optimal or target level of cash holdings; however, there is a delay in adjusting to target or optimal level of cash holdings which is due to the fact that firms do not immediately adjust their cash holdings to an optimal level but take some time because adjustment also entails some costs. The results are consistent with Shah (2011) who found the same behavior of Pakistani firms to adjust to the target level of cash. Earlier, Rehman and Wang (2015) and Rehman et al. (2016) found the same adjustment behavior in Chinese firms and suggested

that Chinese firms follow an optimal or target level of cash holdings and they adjust their Corporate cash cash holdings accordingly. This notion corresponds to a tradeoff model of cash adjustment.

The GMM estimation for overall firms has statistical validity which is tested through three post-estimation tests and other parameters. The number of groups is greater than the number of instruments. The number of groups is 200 and the number of instruments is 52. Sargan test and Abond test are also statistically insignificant having values of 0.329 and 0.385, respectively. Sargan test is for robustness of model and Abond test is to check second-order autocorrelation and it has confirmed the absence of second-order autocorrelation in the model. All the independent variables have also maintained their positive coefficients and statistical significance in GMM regression except net working capital which has a negative coefficient and it is statistically insignificant as well indicating a large incurrence of current liabilities.

4.4 Determinants of corporate cash holdings

Table V also shows the relationship of cash holdings with determinants of cash holdings. Coefficient for firm size is positive and also statistically significant which suggests that firms with higher profits tend to hoard more cash than firms having lower profits because larger firms enjoy economies of scale and large market shares so they keep huge cash reserves. This notion is consistent with some previous studies which are Opler et al. (1999), Shah (2011) and Rehman et al. (2016). Leverage is also positive and statistically significant which is in line with the previous research on tradeoff theory. Firms with high debt to assets ratio tend to keep large cash reserves to cope with bankruptcy risk and financial crisis. Highly levered firms accumulate cash reserves by following precautionary motive of holding cash. The same results were derived by Rehman et al. (2016).

Cash flow has a positive coefficient and it is significant as well which means that firms having huge amounts of cash inflow tend to have larger cash reserves and a large portion of cash flow is reserved as cash to be used as a ready source of liquidity later on. The results are consistent with Ferreira and Vilela (2004) and Shah (2011). Consistent with previous studies including Ozkan and Ozkan (2004), Chen (2008) and Duchin (2010), growth opportunities (GROW) also have a positive and significant sign which indicates that firms having higher growth opportunities also keep larger amounts of cash reserves because their high market-to-book ratio represents more growth opportunities for them and they keep cash reserves to finance their valuable projects. The results are in conformity with the pecking order and tradeoff theory. Net working capital has a positive coefficient yet it is statistically insignificant which means that there is excess of current liabilities incurred by firms and that there is a longer cash conversion cycle as well. For capital expenditure, both the sign and coefficient are positive which is according to the tradeoff theory which holds that firms having high capital expenditure tend to keep large cash in their reserves. This result is in line with Opler et al. (1999) and Rehman et al. (2016).

4.5 Adjustment speed for above and below target firms

Table VI shows regression results for above and below target firms. In case of above target firms, the value adjustment coefficient for lagged cash variable (CASH (L1)) is 0.373 which is positive and statistically significant with t-test value (9.71). For below target firms, the value of adjustment coefficient for lagged cash variable (CASH (L1)) is 0.502 which is also positive and statistically significant with t-test value (35.69). When adjustment coefficients of lagged cash variable are subtracted from 1, we obtained adjustment rate of 0.63 (1-0.37) and 0.5 (1–0.5) for above and below target firms, respectively. The positive and statistically significant lagged coefficient of cash holdings depicts the presence of tradeoff behavior across symmetry.

JABES 26.1

88

				GN	ЛM					
	Above target firms Below target firms									
Variables	Coef.	SE	t-test	<i>p</i> > <i>t</i>	Coef.	SE	t-test	<i>p</i> > <i>t</i>		
CASH										
CASH (L1)	0.373	0.038	9.71	0.000	0.502	0.014	35.69	0.000		
SIZE	0.039	0.032	1.21	0.225	-0.023	0.017	-1.29	0.196		
LEV	-0.019	0.007	-2.47	0.014	0.004	0.002	1.94	0.052		
OCF	0.148	0.02	7.26	0.001	0.04	0.009	4.46	0.001		
GROW	0.009	0.001	9.08	0.001	-0.005	0.004	-1.42	0.156		
NWC	-0.004	0.003	-0.13	0.90	0.027	0.008	3.14	0.002		
CAPEX	0.008	0.018	0.43	0.667	0.004	0.012	0.03	0.973		
_CONS	-0.099	0.161	-0.61	0.539	0.105	0.084	1.26	0.208		
Adj. rate	0.63				0.5					
No. of groups	200				200					
No. of instruments	52				52					
Sargan test	0.395									
Abond test	0.098									

Table VI.GMM regression results for above and below target firms

Notes: GMM is Arellano and Bond estimation. *t*-test values are given in the table. *CASH* is measured by dividing the cash holdings of a firm by total assets. CASH (L1) is the lagged cash variable. Firm size is measured by taking natural logarithm of a firm's total assets. Leverage is measured by dividing total debt of firm by its total assets. Cash flow is measured by dividing net operating cash flows of a firm with total assets. Growth opportunities are measured by dividing market value of equity with book value of equity. Net working capital is measured by subtracting current liabilities from current assets and then dividing the resulting figure with firm's total assets for the purpose of scaling. Capital expenditure is measured by dividing the amount of fixed assets by total assets

The results indicate that downward adjustment speed is higher than upward adjustment speed and the findings are consistent with Rehman *et al.* (2016) who tested and proved the same argument. Thus, the results provide a significant support for acceptance of our hypothesis that downward adjustment speed is higher than upward adjustment speed of corporate cash holdings. The number of groups is greater than the number of instruments for GMM and Sargan and Abond tests have shown insignificant values.

4.6 Financial constraints and adjustment speed of cash holdings

Tables VII and VIII represent results for speed of adjustment of corporate cash holdings with financial constraints. GMM estimation has been given. For financial constraints, three measures have been used. Table VII solely represents results for Altman's Z score measure of financial constraints while Table VIII shows results for SA1 (based on assets to measure financial constraints) and SA2 (based on sales to measure financial constraints). The coefficients of lagged cash variable CASH (L1) are positive and statistically significant across all three measures of financial constraints which is the clear indication of the fact that Pakistani firms tend to follow an optimal or target level of cash holdings in both situations of FC and FUC. Adjustment coefficient which is the coefficient of lagged cash variable (CASH (L1)) for Altman's Z score for FC and FUC firms is 0.085 and 0.43, respectively.

Based on the results of GMM for Altman's Z score measure of financial constraints, adjustment rate is 0.92 (1–0.085) and 0.57 (1–0.43) for FC and FUC firms, respectively. The number of groups is greater than the number of instruments. Sargan test and Abond test are also insignificant.

In Table VIII, adjustment speed for SA1 index is 0.604 (1-0.396) and 0.442 (1-0.558) for FC and FUC firms, respectively. For SA2 index, adjustment speed is 0.73 (1-0.27) and 0.452 (1-0.548) for FC and FUC firms, respectively. Based on the results of above three measures

	Z score					Corporate cash holdings			
		Constrain	ed firms				firms		noidings
Variables	Coef.	SE	t-test	<i>p</i> > <i>t</i>	Coef.	SE	t-test	<i>p</i> > <i>t</i>	
CASH									
CASH (L1)	0.085	0.022	8.71	0.000	0.43	0.019	21.91	0.000	
SIZE	-0.01	0.028	-5.39	0.001	-0.022	0.029	-0.75	0.46	
LEV	0.001	0.007	0.22	0.824	0.041	0.008	4.87	0.001	89
OCF	0.024	0.022	12.78	0.001	0.158	0.02	7.89	0.001	
GROW	0.001	0.001	2.82	0.005	0.001	0.001	2.53	0.001	
NWC	0.001	0.031	2.26	0.024	0.031	0.028	1.100	0.27	
CAPEX	-0.001	0.032	-0.47	0.642	-0.064	0.035	-1.79	0.07	
_CONS	0.072	0.145	5.87	0.001	0.114	0.15	0.76	0.45	
Adj. rate	0.92				0.57				
No. of groups	122				125				
No. of instruments	52				52				
Sargan test	0.55				0.1				
Abond test	0.79				0.38				

Notes: *t*-test values are given in the table. GMM is Arellano and Bond estimation. *Z* score is Altman's *Z* score. *CASH* is measured by dividing the cash holdings of a firm by total assets. CASH (L1) is the lagged cash variable. Firm size is measured by taking natural logarithm of a firm's total assets. Leverage is measured by dividing total debt of firm by its total assets. Cash flow is measured by dividing net operating cash flows of a firm with total assets. Growth opportunities are measured by dividing market value of equity with book value of equity. Net working capital is measured by subtracting current liabilities from current assets and then dividing the resulting figure with firm's total assets for the purpose of scaling. Capital expenditure is measured by dividing the amount of fixed assets by total assets

Table VII.
Altman Z score's
regression results for
FC and FUC
firms (GMM)

of financial constraints, there is no considerable evidence in support of our second hypothesis that FUC firms have higher adjustment speed for cash holdings than FC firms.

According to our findings, FC firms move more quickly toward optimal level of cash holdings than FUC firms in case of deviation from the target level of cash. The results are consistent with Rashid and Ashfaq (2017) and Han and Qiu (2007) who also found a higher tendency of accumulating cash by FC firms than FUC firms. Higher levels of cash holdings in FC firms are associated with higher investment and hedging needs. Furthermore, large cash reserves allow FC firms to undertake certain profitable investments which might otherwise be ignored. FC firms also hoard cash to avoid costly external financing. The model estimations for all the measures of financial constraints are statistically significant because all have shown more number of groups than number of instruments. Sargan test is also insignificant to prove that the results are robust. Abond test is also insignificant which shows the absence of second-order autocorrelation.

4.7 Downward and upward adjustment speed across financial constraints

Table IX corresponds to GMM regression results for asymmetric adjustment speed of corporate cash holdings to the target level while incorporating firm's financial constraints. Firm-level observations above the optimal or target level of cash holdings are given in the first three columns of Table IX while observations below the optimal level are presented in the last three columns of Table IX. Panel A represents Altman's Z score measure of financial constraints. Panel B corresponds to SA1 measure and Panel C is for SA2 measure of financial constraints. In Panel A for Altman's Z score, downward adjustment speed for above target firms is 0.77 and 0.99 for FC and FUC firms, respectively. Upward adjustment speed for below target firms is 0.81 and 0.54 for FC and FUC firms, respectively. It indicates that downward adjustment speed is higher only with FUC firms and upward adjustment



JABES 26,1

90

		Constrai	ned firms			FUC 1	firms	
Variables	Coef.	SE	t-test	<i>p</i> > <i>t</i>	Coef.	SE	t-test	<i>p</i> > <i>t</i>
SA1 index								
CASH								
CASH (L1)	0.396	0.000	841.7	0.000	0.558	0.014	39.86	0.000
SIZE	-0.172	0.001	-101.5	0.001	0.042	0.013	3.08	0.002
LEV	0.001	0.001	-0.90	0.346	0.021	0.003	5.44	0.001
OCF	0.001	0.001	208.7	0.001	0.163	0.016	9.67	0.001
GROW	0.001	0.001	85.5	0.001	0.001	0.001	-0.19	0.85
NWC	0.002	0.001	30.4	0.001	0.13	0.02	6.73	0.001
CAPEX	-0.03	0.001	-72.4	0.001	-0.036	0.013	-2.72	0.007
_CONS	0.65	0.001	66.2	0.001	-0.22	0.078	-2.89	0.004
Adj. rate	0.604				0.442	0.986		
No. of groups	62				92			
No. of instruments	52				52			
Sargan test	0.21				0.37			
Abond test	0.32				0.49			
SA2 index								
CASH (L1)	0.27	0.000	336.12	0.000	0.548	0.016	33.99	0.000
SIZE	-0.09	0.001	-61.3	0.001	-0.03	0.011	-2.77	0.006
OCF	0.037	0.001	88.16	0.001	0.142	0.012	11.01	0.001
GROW	0.001	0.001	26.97	0.001	0.001	0.001	2.58	0.01
NWC	0.001	0.001	-30.78	0.001	0.039	0.026	1.5	0.134
CAPEX	-0.01	0.001	-4.84	0.001	-0.102	0.011	-8.77	0.001
_CONS	0.377	0.001	44.6	0.001	0.187	0.062	2.99	0.003
Adj. rate	0.73				0.452			
No. of groups	79				92			
No. of instruments	52				52			
Sargan test	0.44				0.31			
Abond test	0.61				0.07			

Notes: *t*-test values are given in the table. GMM is Arellano and Bond estimation. SA1 is assets-based measure of financial constraints. SA2 is sales-based measure of financial constraints. Cash is measured by dividing the cash holdings of a firm by total assets. CASH (L1) is the lagged cash variable. Firm size is measurement made by taking natural logarithm of a firm's total assets. Leverage is measured by dividing total debt of firm by its total assets. Cash flow is measured by dividing net operating cash flows of a firm with total assets. Growth opportunities are measured by dividing market value of equity with book value of equity. Net working capital is measured by subtracting current liabilities from current assets and then dividing the resulting figure with firm's total assets for the purpose of scaling. Capital expenditure is measured by dividing the amount of fixed assets by total assets

Table VIII.SA index's regression results for FC and FUC firms (GMM)

rate is higher only for FC firms. These are mixed results with no clear indication that downward adjustment speed is higher even after financial constraints are controlled.

Higher downward adjustment speed for FUC firms is due to excess cash holdings by these firms and higher upward speed for FC firms is because of the reason that these firms hold more cash to cope with cash flow volatility and financial crisis. Based upon GMM estimates for Altman's *Z* score, the results are not consistent with our third hypothesis. According to SA1 measure in Panel B, downward adjustment speed for FC and FUC firms is 0.58 and 0.47, respectively. For below target firms, this speed is 0.61 and 0.58 for FC and FUC firms, respectively. It indicates that upward adjustment speed is higher than downward adjustment speed after controlling for financial constraints. Furthermore, while analyzing SA2 measure in Panel C, downward adjustment speed is 0.67 and 0.67 for FC and FUC firms, respectively, and for below target firms, adjustment speed is 0.6 and 0.56 for FC and FUC firms, respectively. It indicates that downward adjustment speed is higher than

-	Above	e target	Below	r target	Corporate cash
	Constrained	FUC	Constrained	FUC	holdings
Panel A: ZSCORE					
Adj. rate	0.77	0.992	0.81	0.54	
CASH (L1)	-0.23***(-14.66)	-0.008*(-0.49)	0.19*** (14.34)	0.46*** (33.65)	
No. of groups	27	67	114	100	0.1
No. of instruments	44	52	52	52	91
Abond test	0.27	0.59	0.5	0.04	
Panel B: SA1					
Adj. rate	0.58	0.47	0.61	0.58	
CASH (L1)	0.42*** (11.61)	0.53*** (64.21)	0.39*** (1420)	0.42*** (53.43)	
No. of groups	26	45	57	81	
No. of instruments	51	52	52	52	
Abond test	0.9	0.66	0.13	0.39	
Panel C: SA2					
Adj. rate	0.67	0.67	0.6	0.56	
CASH (L1)	0.33*** (-11.16)	0.33***(-14.65)	0.4***(-1628)	0.44*** (-43.15)	
No. of groups	30	45	70	84	
No. of instruments	49	52	52	52	
Abond test	0.69	0.13	0.34	0.33	
37		7 . 41. 1 .	7 011 .	1 1 6	

Notes: *t*-test values are given in the table. *Z* score is Altman's *Z* score. SA1 is assets-based measure of financial constraints. SA2 is sales-based measure of financial constraints. GMM is Arellano and Bond estimation. Cash is calculated by dividing the cash holdings of a firm by total assets. CASH (L1) is the lagged cash variable. ****Statistical level of significance at 90 and 99 percent, respectively

Table IX.
Regression results
for asymmetric speed
and constraints

upward adjustment speed even after controlling for financial constraints. The result is consistent with the findings of Rehman $et\ al.\ (2016)$ who reported the same results for SA2 measure of financial constraints yet they presented the same findings for SA1 measure and for Altman's Z score as well.

For firms to move downward or to adjust to their optimal or target level of cash holdings, it is quite easy because they can pay taxes, dividends and make investments in profitable ventures to cut down their excess cash holdings. Although the results of SA2 measure of financial constraints support our third hypothesis, Altman's Z score and SA1 measure did not provide any substantial support to this hypothesis that is why it is rejected.

5. Conclusion

The main focus of this study is to find out the upward and downward adjustment behavior of Pakistani firms toward their optimal or target level of cash holdings. Prior research studies of capital structure (Drobetz and Wanzenried, 2006; Almeida *et al.*, 2004; Han and Qiu, 2007; Al-Najjar, 2013) have been followed to understand the tendency of firms to opt for a target level of cash and for this purpose, firms have been divided into above and below target firms. For the estimation of adjustment speed for above and below target firms, Arellano and Bond (GMM) estimator has been used which is a dynamic model for panel data regression and is suitable to analyze the speed of adjustment of firms. On the first stance, the results prove that downward speed of adjustment is higher than upward speed of adjustment and it is because of the reason that firms with cash holdings above target level can easily cut down their cash reserves either by making necessary debt payments or by investing in profitable projects. Financial constraints have also been incorporated in our research model to check for the adjustment speed of FC and FUC firms and to prove that the downward speed of adjustment still remains higher even after controlling for financial



constraints. But when it comes up with financial constraints, there is not any considerable evidence that this speed of adjustment holds after controlling for financial constraints. Furthermore, this study uses three measures of financial constraints (i.e. Altman's Z score, SA1 index and SA2 index) to explore the adjustment speed of corporate cash holdings but all three measures failed to provide any substantial support to our hypotheses.

The results are mixed as at the first place, FC firms have appeared to adjust more speedily then FUC firms and second, there is no evidence of higher rate of downward adjustment of cash holdings after controlling for financial constraints. The results are consistent with Rashid and Ashfaq (2017) who also found a positive association between corporate cash holdings and financial constraints. But for asymmetric adjustment of cash holdings across financial constraints, there is no evidence from prior literature which gives the same results.

This study fails to conciliate financial constraints to address the adjustment speed of corporate cash holdings and it is because of the fact that Pakistani firms do not adjust promptly to their optimal level when they are FC as they do not have certain valuable assets in their portfolio which can be offered as collateral to back the debt service and they may also run out of cheap debt obligations, they are mainly affected by information asymmetry, they ought to borrow at high costs and their small size, young age, lower level of income also make them to hoard more cash but still they do not adjust to the target level of cash holdings. Moreover, Pakistani firms with high levels of cash holdings are prone to financial distress (Afza and Adnan, 2007; Kruja and Borici, 2016), suffer more from cash flow volatility and spend huge amounts of cash on research and development. These factors also restrict their downward adjustment speed toward target level of cash.

There are no alternatives available to Pakistani firms for their downward adjustment of cash holdings when financial constraints are involved. There are more adjustment costs associated with downward adjustment than for upward adjustment of cash holdings. For example, Pakistani firms may have their debts matured and payable so they need to pay them first instead of using cash immediately, or they may have dividends outstanding so they need to make dividends payments immediately. These adjustment costs make it difficult for Pakistani firms to adjust quickly to their optimal level of cash. Moreover, high dividend payout ratio and firm's age also hinder the speedy adjustment of corporate cash holdings in Pakistani firms because these factors are considered as internal financial constraints (Azam and Shah, 2011), and costly external financing is also a hurdle for firms to adjust quickly to their optimal or target level of cash. Moreover, transaction costs, opportunity costs, agency costs of financial upsets and high investment costs are other important factors responsible for lower downward speed of adjustment of Pakistani firms (Azmat, 2014).

The results are specifically useful for managers, policymakers, investors and researchers. Cash holdings policies can be revised according to the results of this study and to understand the adjustment behavior of FC and FUC firms toward an optimal or target level of cash. As FC firms appear to hoard more cash in their reserves, these findings are particularly helpful for policymakers that if they want to reduce the intensity of financial constraints, they must take steps to reduce barriers to inter-financial markets and take initiatives to improve the functioning of overall capital markets.

References

Afza, T. and Adnan, S.M. (2007), "Determinants of corporate cash holdings: a case study of Pakistan", Proceedings of Singapore Economic Review Conference (SERC), Vol. 2007, Singapore, August.

Almeida, H., Campello, M. and Weisbach, M.S. (2004), "The cash flow sensitivity of cash", *The Journal of Finance*, Vol. 59 No. 4, pp. 1777-1804.

93

holdings

- Al-Najjar, B. (2013), "The financial determinants of corporate cash holdings: evidence from some Corporate cash emerging markets", International Business Review, Vol. 22 No. 1, pp. 77-88.
- Arellano, M. and Bond, S. (1991), "Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations", The Review of Economic Studies, Vol. 58 No. 2, pp. 277-297.
- Azam, M. and Shah, S.A. (2011), "Internal financial constraints, external financial constraints and investment choice: evidence from Pakistani firms", Australian Journal of Business and Management Research, Vol. 1 No. 8, pp. 18-22.
- Azmat, Q.U.A. (2014), "Firm value and optimal cash level: evidence from Pakistan", International Journal of Emerging Markets, Vol. 9 No. 4, pp. 488-504.
- Bancel, F. and Mittoo, U.R. (2011), "Financial flexibility and the impact of global financial crisis: evidence from France", International Journal of Managerial Finance, Vol. 7 No. 2, pp. 179-216.
- Bates, T.W., Kahle, K.M. and Stulz, R.M. (2009), "Why do us firms hold so much more cash than they used to?", The Journal of Finance, Vol. 64 No. 5, pp. 1985-2021.
- Campello, M., Graham, J.R. and Harvey, C.R. (2010), "The real effects of financial constraints: evidence from a financial crisis", Journal of Financial Economics, Vol. 97 No. 3, pp. 470-487.
- Chang, Y., Benson, K.L. and Faff, R.W. (2017), "Asymmetric modelling of the adjustment speed of cash holdings", Asian Finance Association (AsianFA) 2017 Conference, February, available at: https://ssrn.com/abstract=2914986; https://dx.doi.org/10.2139/ssrn.2914986.
- Chen, T., Harford, J. and Lin, C. (2017), "Financial flexibility and corporate cash policy abstract: debt capacity creates financial flexibility and collateral-based debt capacity is the least sensitive to cash flow shocks", Working Papers No. 052017, Hong Kong Institute for Monetary Research, available at: https://ideas.repec.org/p/hkm/wpaper/052017.html.
- Chen, Y.R. (2008), "Corporate governance and cash holdings: listed new economy versus old economy firms", Corporate Governance: An International Review, Vol. 16 No. 5, pp. 430-442.
- DeAngelo, H., DeAngelo, L. and Whited, T.M. (2011), "Capital structure dynamics and transitory debt", Journal of Financial Economics, Vol. 99 No. 2, pp. 235-261, available at: https://doi.org/10.1016/j. ifineco.2010.09.005
- Denis, D.J. (2011), "Financial flexibility and corporate liquidity", Journal of Corporate Finance, Vol. 17 No. 3, pp. 667-674.
- Denis, D.J. and McKeon, S.B. (2012), "Debt financing and financial flexibility evidence from proactive leverage increases", The Review of Financial Studies, Vol. 25 No. 6, pp. 1897-1929.
- Denis, D.J. and Sibilkov, V. (2009), "Financial constraints, investment, and the value of cash holdings", The Review of Financial Studies, Vol. 23 No. 1, pp. 247-269.
- Denis, D.J. and Sibilkov, V. (2010), "Financial constraints, investment, and the value of cash holdings", Review of Financial Studies, Vol. 23 No. 1, pp. 247-269.
- Dittmar, A., Mahrt-Smith, J. and Servaes, H. (2003), "International corporate governance and corporate cash holdings", Journal of Financial and Quantitative Analysis, Vol. 38 No. 1, pp. 111-133.
- Drobetz, W. and Wanzenried, G. (2006), "What determines the speed of adjustment to the target capital structure?", Applied Financial Economics, Vol. 16 No. 13, pp. 941-958, available at: https://doi.org/ 10.1080/09603100500426358
- Duchin, R. (2010), "Cash holdings and corporate diversification", Journal of Finance, Vol. 65 No. 3, pp. 955-992.
- Ferreira, M.A. and Vilela, A.S. (2004), "Why do firms hold cash? Evidence from EMU countries", European Financial Management, Vol. 10 No. 2, pp. 295-319.
- Foley, C.F., Hartzell, J.C., Titman, S. and Twite, G. (2007), "Why do firms hold so much cash? A tax-based explanation", Journal of Financial Economics, Vol. 86 No. 3, pp. 579-607.
- Gamba, A. and Triantis, A. (2008), "The value of financial flexibility", The Journal of Finance, Vol. 63 No. 5, pp. 2263-2296.



- García-Teruel, P.J. and Martínez-Solano, P. (2008), "On the determinants of SME cash holdings: evidence from Spain", *Journal of Business Finance & Accounting*, Vol. 35 Nos 1-2, pp. 127-149.
- Graham, J.R. and Harvey, C.R. (2001), "The theory and practice of corporate finance: evidence from the field", *Journal of Financial Economics*, Vol. 60 Nos 2-3, pp. 187-243.
- Hadlock, C. and Pierce, J. (2010), "New evidence on measuring financial constraints: moving beyond the KZ index", *Review of Financial Studies*, Vol. 23 No. 5, pp. 1909-1940.
- Han, S. and Qiu, J. (2007), "Corporate precautionary cash holdings", Journal of Corporate Finance, Vol. 13 No. 1, pp. 43-57.
- Harford, J., Klasa, S. and Maxwell, W.F. (2014), "Refinancing risk and cash holdings", The Journal of Finance, Vol. 69 No. 3, pp. 975-1012.
- Harford, J., Li, K. and Zhao, X. (2008), "Corporate boards and the leverage and debt maturity choices", International Journal of Corporate Governance, Vol. 1 No. 1, pp. 3-27.
- Harford, J., Mansi, S. and Maxwell, W.F. (2008), "Corporate governance and firm cash holdings in the US", *Journal of Financial Economics*, No. 3.
- Hovakimian, A., Opler, T. and Titman, S. (2001), "The debt equity choice", *Journal of Financial and Quantitative Analysis*, Vol. 36 No. 1, pp. 1-24.
- Jarrow, R., Krishenik, A. and Minca, A. (2018), "Optimal cash holdings under heterogeneous beliefs", Mathematical Finance, Vol. 28 No. 2, pp. 712-747.
- Jensen, M.C. (1986), "Agency costs of free cash flow, corporate finance, and takeovers", *The American Economic Review*, Vol. 76 No. 2, pp. 323-329.
- Jiang, Z. and Lie, E. (2016), "Cash holding adjustments and managerial entrenchment", Journal of Corporate Finance, Vol. 36, pp. 190-205.
- John, T.A. (1993), "Accounting measures of corporate liquidity, leverage, and costs of financial distress", Financial Management, Vol. 22 No. 3, pp. 91-100.
- Keynes, J.M. (1936), "The general theory of employment, investment, and money", London and New York, NY, available at: https://sites.google.com/site/biblioeconomicus/KeynesJohnMaynard-TheGeneralTheoryOf EmploymentInterestAndMoney.pdf (accessed January 7, 2014).
- Kruja, A. and Borici, A. (2016), "Determinants of firm's cash holding", International Journal of Economics, Commerce and Management, Vol. 4 No. 4, pp. 41-52.
- Lozano, M.B. and Durán, R.F. (2017), "Family control and adjustment to the optimal level of cash holding", *The European Journal of Finance*, Vol. 23 No. 3, pp. 266-295.
- Martínez-Sola, C., García-Teruel, P.J. and Martínez-Solano, P. (2013), "Corporate cash holding and firm value", *Applied Economics*, Vol. 45 No. 2, pp. 161-170.
- Mikkelson, W.H. and Partch, M.M. (2003), "Do persistent large cash reserves hinder performance?", Journal of Financial and Quantitative Analysis, Vol. 38 No. 2, pp. 275-294.
- Miller, M.H. and Orr, D. (1966), "A model of the demand for money by firms", *The Quarterly Journal of Economics*, Vol. 80 No. 3, pp. 413-435.
- Modigliani, F. and Miller, M.H. (1958), "The cost of capital, corporation finance and the theory of investment", The American Economic Review, Vol. 48 No. 3, pp. 261-297.
- Mulligan, C.B. (1997), "Scale economies, the value of time, and the demand for money: longitudinal evidence from firms", *Journal of Political Economy*, Vol. 105 No. 5, pp. 1061-1079.
- Myers, S.C. (1984), "The capital structure puzzle", The Journal of Finance, Vol. 39 No. 3, pp. 574-592.
- Nikolov, B. and Whited, T.M. (2014), "Agency conflicts and cash: estimates from a dynamic model", The Journal of Finance, Vol. 69 No. 5, pp. 1883-1921.
- Opler, T., Pinkowitz, L., Stulz, R. and Williamson, R. (1999), "The determinants and implications of corporate cash holdings", *Journal of Financial Economics*, Vol. 52 No. 1, pp. 3-46.
- Ozkan, A. and Ozkan, N. (2004), "Corporate cash holdings: an empirical investigation of UK companies", *Journal of Banking & Finance*, Vol. 28 No. 9, pp. 2103-2134.

- Rashid, A. and Ashfaq, M. (2017), "Financial constraints and corporate cash holdings: an empirical Corporate cash analysis using firm level data", *Annals of Financial Economics*, Vol. 12 No. 2, p. 1750009.
- Rehman, A., Wang, M. and Kabiraj, S. (2016), "Dynamics of corporate cash holdings in Chinese firms: an empirical investigation of asymmetric adjustment rate and financial constraints", *Asian Academy of Management Journal of Accounting & Finance*, Vol. 12 No. 2, pp. 127-152.
- Rehman, A.U. and Wang, M. (2015), "Corporate cash holdings and adjustment behavior in Chinese firms: an empirical analysis using generalized method of moments", Australasian Accounting Business & Finance Journal, Vol. 9 No. 4, pp. 20-37.
- Shah, A. (2011), "The corporate cash holdings: determinants and implications", African Journal of Business Management, Vol. 5, No. 34, pp. 12939-12950, available at: https://ssrn.com/abstract=1982639
- Uyar, A. and Kuzey, C. (2014), "Determinants of corporate cash holdings: evidence from the emerging market of Turkey", Applied Economics, Vol. 46 No. 9, pp. 1035-1048.
- Venkiteshwaran, V. (2011), "Partial adjustment toward optimal cash holding levels", *Review of Financial Economics*, Vol. 20 No. 3, pp. 113-121.
- Xie, X., Wang, Y., Zhao, G. and Lu, W. (2017), "Cash holdings between public and private insurers-a partial adjustment approach", *Journal of Banking & Finance*, Vol. 82, pp. 80-97.

Further reading

- Acharya, V.V., Schnabl, P. and Suarez, G. (2013), "Securitization without risk transfer", *Journal of Financial Economics*, Vol. 107 No. 3, pp. 515-536.
- Ahsan, S. and Ullah, N. (2013), "Impact of cashflow volatility on cash-cash flow sensitivity of Pakistani firms", *Journal of Business and Management*, Vol. 8 No. 1, pp. 85-97.
- Ali, S., Ullah, M. and Ullah, N. (2016), "Determinants of corporate cash holdings: 'a case of textile sector in Pakistan", available at: https://ssrn.com/abstract=2728200; https://dx.doi.org/10.2 139/ssrn.2728200
- Al-Najjar, B. and Belghitar, Y. (2011), "Corporate cash holdings and dividend payments: evidence from simultaneous analysis", Managerial and Decision Economics, Vol. 32 No. 4, pp. 231-241.
- Anjum, S. and Malik, Q.A. (2013), "Determinants of corporate liquidity an analysis of cash holdings", Journal of Business and Management, Vol. 7 No. 2, pp. 94-100.
- Arfan, M., Basri, H., Handayani, R., Majid, M.S.A., Fahlevi, H. and Dianah, A. (2017), "Determinants of cash holding of listed manufacturing companies in the Indonesian Stock Exchange", DLSU Business & Economics Review, Vol. 26 No. 2, pp. 1-12.
- Azmat, Q.U.A. and Iqbal, A.M. (2017), "The role of financial constraints on precautionary cash holdings: evidence from Pakistan", *Economic Research (Ekonomska Istraživanja)*, Vol. 30 No. 1, pp. 596-610.
- Beck, T., Demirgüç-Kunt, A., Laeven, L. and Maksimovic, V. (2006), "The determinants of financing obstacles", *Journal of International Money and Finance*, Vol. 25 No. 6, pp. 932-952.
- Bigelli, M. and Sánchez-Vidal, J. (2012), "Cash holdings in private firms", Journal of Banking & Finance, Vol. 36 No. 1, pp. 26-35.
- Bliss, B.A., Cheng, Y. and Denis, D.J. (2015), "Corporate payout, cash retention, and the supply of credit: evidence from the 2008–2009 credit crisis", *Journal of Financial Economics*, Vol. 115 No. 3, pp. 521-540.
- Blundell, R. and Bond, S. (2000), "GMM estimation with persistent panel data: an application to production functions", *Econometric Reviews*, Vol. 19 No. 3, pp. 321-340.
- Chan, H.W., Lu, Y. and Zhang, H.F. (2013), "The effect of financial constraints, investment policy, product market competition and corporate governance on the value of cash holdings", Accounting & Finance, Vol. 53 No. 2, pp. 339-366.
- Cheema, A. (2003), "Corporate governance in Pakistan: issues and concerns", *The Journal*, Vol. 8 No. 2, pp. 7-19.



- Deloof, M. (2003), "Does working capital management affect profitability of Belgian firms?", *Journal of Business Finance & Accounting*, Vol. 30 Nos 3-4, pp. 573-588.
- D'Mello, R., Krishnaswami, S. and Larkin, P.J. (2008), "Determinants of corporate cash holdings: evidence from spin-offs", *Journal of Banking & Finance*, Vol. 32 No. 7, pp. 1209-1220.
- Drobetz, W. and Grüninger, M.C. (2007), "Corporate cash holdings: evidence from Switzerland", Financial Markets and Portfolio Management, Vol. 21 No. 3, pp. 293-324.
- Faulkender, M. and Wang, R. (2006), "Corporate financial policy and the value of cash", The Journal of Finance, Vol. 61 No. 4, pp. 1957-1990.
- Fresard, L. (2010), "Financial strength and product market behavior: the real effects of corporate cash holdings", *The Journal of Finance*, Vol. 65 No. 3, pp. 1097-1122.
- Gao, H., Harford, J. and Li, K. (2013), "Determinants of corporate cash policy: insights from private firms", *Journal of Financial Economics*, Vol. 109 No. 3, pp. 623-639.
- Gill, A. and Shah, C. (2012), "Determinants of corporate cash holdings: evidence from Canada", International Journal of Economics and Finance, Vol. 4 No. 1, pp. 70-79.
- Guney, Y., Ozkan, A. and Ozkan, N. (2007), "International evidence on the non-linear impact of leverage on corporate cash holdings", *Journal of Multinational Financial Management*, Vol. 17 No. 1, pp. 45-60.
- Hardin, W.G., Highfield, M.J., Hill, M.D. and Kelly, G.W. (2009), "The determinants of REIT cash holdings", *The Journal of Real Estate Finance and Economics*, Vol. 39 No. 1, pp. 39-57.
- Harford, J. (1999), "Corporate cash reserves and acquisitions", The Journal of Finance, Vol. 54 No. 6, pp. 1969-1997.
- Harris, M. and Raviv, A. (1990), "Capital structure and the informational role of debt", *The Journal of Finance*, Vol. 45 No. 2, pp. 321-349.
- Hilgen, M.H. (2015), "The determinants of cash holdings: evidence from German listed firms", master's thesis, University of Twente, Enschede, available at: www.bing.com/search?q=Enschede +wikipedia&FORM=LFACTRE; www.bing.com/search?q=Netherlands+wikipedia&FORM=LFACTRE
- Hill, M.D., Kelly, G.W. and Highfield, M.J. (2010), "Net operating working capital behavior: a first look", Financial Management, Vol. 39 No. 2, pp. 783-805.
- Hsu, P.H., Li, F. and Lin, T.C. (2016), "Innovative firms hold more cash? the international evidence", working paper, University of Hong Kong.
- Javed, A.Y. and Iqbal, R. (2006), "Corporate governance and firm performance: evidence from Karachi stock exchange", The Pakistan Development Review, Vol. 45, pp. 947-964.
- Javed, A.Y. and Iqbal, R. (2007), "External financial resource management by listed Pakistani firms", The Pakistan Development Review, Vol. 46, pp. 449-464.
- Jinkar, R.T. (2013), "Analisa Faktor-faktor Penentu Kebijakan cash holding perusahaan Manufaktur di Indonesia", Jurnal Edisi, Vol. 42, pp. 126-129.
- Jung, K., Kim, Y.C. and Stulz, R. (1996), "Timing, investment opportunities, managerial discretion, and the security issue decision", *Journal of Financial Economics*, Vol. 42 No. 2, pp. 159-186.
- Khurana, I.K., Martin, X. and Pereira, R. (2006), "Financial development and the cash flow sensitivity of cash", *Journal of Financial and Quantitative Analysis*, Vol. 41 No. 4, pp. 787-808.
- Kim, M., Shin, Y. and Dang, V.A. (2009), "Asymmetric capital structure adjustments: new evidence from dynamic panel threshold models", SSRN working paper, available at: http://ssrn.com/ abstract=1444488
- La Cava, G. and Windsor, C. (2016), "Why do companies hold cash?", No. rdp2016-03, Reserve Bank of Australia, Sydney.
- Love, I. (2003), "Financial development and financing constraints: international evidence from the structural investment model", *The Review of Financial Studies*, Vol. 16 No. 3, pp. 765-791.

holdings

- Mirza, H.H. and Azfa, T. (2010), "Ownership structure and cash flows as determinants of corporate Corporate cash dividend policy in Pakistan", International Business Research, Vol. 3 No. 3, pp. 210-221, available at: https://ssrn.com/abstract=2019549
- Nafees, B., Ahmad, N. and Rasheed, A. (2017), "The determinants of cash holdings: evidence from SMEs in Pakistan", Paradigms, Vol. 11 No. 1, pp. 111-116.
- Ogundipe, S.E., Idowu, A. and Ogundipe, L.O. (2012), "Working capital management, firms' performance and market valuation in Nigeria", World Academy of Science, Engineering and Technology, Vol. 61 No. 1, pp. 1196-1200.
- Rajan, R.G. and Zingales, L. (1995), "What do we know about capital structure? Some evidence from international data", The Journal of Finance, Vol. 50 No. 5, pp. 1421-1460.
- Rizwan, M.F. (2015), "Determinants of corporate cash holdings and its implications; evidence from Pakistan's corporate sector", doctoral dissertation, Mohammad Ali Jinnah University, Islamabad.
- Rizwan, M.F. and Javed, T. (2011), "Determinants of corporate cash holdings: evidence from Pakistani public sector", Economics, Management and Financial Markets, Vol. 6 No. 1, pp. 344-358.
- Schwetzler, B. and Reimund, C. (2004), "Valuation effect of cash holdings: evidence from Germany", working paper, Jahnallee.
- Subramaniam, V., Tang, T.T., Yue, H. and Zhou, X. (2011), "Firm structure and corporate cash holdings", Journal of Corporate Finance, Vol. 17 No. 3, pp. 759-773.
- Titman, S. and Wessels, R. (1988), "The determinants of capital structure choice", The Journal of Finance, Vol. 43 No. 1, pp. 1-19.
- Ullah, S. and Kamal, Y. (2018), "Corporate cash holdings and shareholder wealth: evidence from Pakistani market". Pakistan Business Review, Vol. 19 No. 4, pp. 978-994.
- Wang, Y., Ji, Y., Chen, X. and Song, C. (2014), "Inflation, operating cycle, and cash holdings", China Journal of Accounting Research, Vol. 7 No. 4, pp. 263-276.
- Wasiuzzaman, S. (2014), "Analysis of corporate cash holdings of firms in Malaysia", Journal of Asia Business Studies, Vol. 8 No. 2, pp. 118-135.
- William, W. and Fauzi, S. (2013), "Analisis pengaruh growth opportunity, net working capital, dan conversion cycle terhadap cash holding perusahaan sektor pertambangan (Analysis of the effect of growth opportunity, net working capital and conversion cycle on corporate cash holding from mining sector)", Jurnal Ekonomi & Keuangan, Vol. 1 No. 2, pp. 72-90.
- Williamson, O. (1988), "Corporate finance and corporate governance", Journal of Finance, Vol. 43, pp. 567-591.
- Wu, Q. and Shamsuddin, A. (2012), "Do industries lead the stock market in Australia? An examination of gradual information diffusion hypothesis", 23rd Australasian Finance and Banking Conference, Sydney, December 15-17.
- Zaidi, R. and Aslam, A. (2005), "Managerial efficiency in family owned firms in Pakistan an examination of listed firms", CMER Working Paper Series No. 06-51, LUMS, Lahore, available at: www.eaber.org/node/22283 (accessed November 10, 2015).

Corresponding author

Ajid ur Rehman can be contacted at: ajid.rehman@gmail.com

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: **permissions@emeraldinsight.com**



© Ghulam Ayehsa Siddiqua, Ajid ur Rehman and Shahzad Hussain. This work is published under (the "License"). Notwithstanding the ProQuest Terms and Conditions, you may use this content in accordance with the terms of the License. https://creativecommons.org/licenses/by-nc/3.0/legalcode